Author's response to reviews

Title: Modulation of apoptosis in human hepatocellular carcinoma (HepG2 cells) by a standardized herbal decoction with anti-hepatocarcinogenic effects.

Authors:

Sameera R Samarakoon (sam@ibmbb.cmb.ac.lk)
Ira Thabrew (irathab@hotmail.com)
Prasanna B Galhena (prasanna@kl.ac.lk)
Kamani H Tennekoon (kamani@ibmbb.cmb.ac.lk)

Version: 2 Date: 27 October 2011

Author's response to reviews: see over
To,
The Editor.
BMC Complementary and Alternative Medicine,

Sub: Submission of Manuscript for publication

Dear Sir,

We intend to publish an article entitled “Modulation of apoptosis in human hepatocellular carcinoma (HepG2 cells) by a standardized herbal decoction with anti-hepatocarcinogenic effects.” in your esteemed journal. We have decided that this manuscript is more suitable for your journal and it is relevant to scope of the journal. On behalf of all the contributors I will act and guarantor and will correspond with the journal from this point onward. All authors have read and agreed to your editorial and peer review policies.

Conflicts of interest: None

We hereby transfer, assign, or otherwise convey all copyright ownership, including any and all rights incidental there to, exclusively to the journal, in the event that such work is published by the journal.

Thanking you,

Yours’ sincerely,

Signature

Sameera R. Samarakoon (Corresponding contributor)
Institute of Biochemistry, Molecular Biology & Biotechnology,
University of Colombo, Cumarathunga Munidasa Mawatha, 90,
Colombo 3, Sri Lanka.

e-mail- ranganath148@yahoo.com, sam@ibmbb.cmb.ac.lk